Author's response to reviews

Title: miR-221 Facilitates the TGFbeta1-Induced Epithelial-Mesenchymal Transition in Human Bladder Cancer Cells by Targeting STMN1

Authors:

Jun Liu (liujun2010_good@163.com)
Jian Cao (caojian356288404@sina.com)
Xiaokun Zhao (xiaokunzhao2014@163.com)

Version: 4 Date: 25 March 2015

Author's response to reviews: see over
Dear reviewer:
I am very grateful to your comments for the manuscript entitled “MS: 1829324592150159
miR-221 Facilitates the TGFbeta1-Induced Epithelial-Mesenchymal Transition in
Human Bladder Cancer Cells by Targeting STMN1”. According with your advice, we
carefully revised our manuscript. Now I answer the questions one-by-one.

EDITORIAL REQUEST:

1. Abbreviations need to be moved to after the Conclusions.
   Answer: Thank you for the suggestion. The abbreviations have been moved to after the
   Conclusions.

2. Introduction needs to be re-named Background.
   Answer: Thank you for the suggestion. “Introduction” was instead by “Background”.

3. Format Authors' Contribution to use author initials rather than full author
   names.
   Answer: Thank you for the suggestion. I have done it as your suggest.

4. Include Acknowledgements with details of any funding.
   Answer: There is no fundings.

5. Include where the RT4 and T24 cell lines were obtained from.
   Answer: I have done it as your suggest.

Most sincerely,
Xiaokun Zhao
Xiaokun Zhao, PhD.
Department of Urology, 2nd Xingya Hospital, Central South University, NO.139
Middle Renmin Road, Changsha, Hunan, China 410011
E-mail: ZXK652348@163.com